

# Northwestern Mediterranean Heavy Precipitation Events in a Warmer Climate: Robust Versus Uncertain Changes With a Large Convection-Permitting Model Ensemble

Cécile Caillaud, Samuel Somot, Hervé Douville, Antoinette Alias, Sophie Bastin, Susanne Brienen, Marie-estelle Demory, Andreas Dobler, Hendrik Feldmann, Thomas Frisius, et al.

# ▶ To cite this version:

Cécile Caillaud, Samuel Somot, Hervé Douville, Antoinette Alias, Sophie Bastin, et al.. Northwestern Mediterranean Heavy Precipitation Events in a Warmer Climate: Robust Versus Uncertain Changes With a Large Convection-Permitting Model Ensemble. Geophysical Research Letters, 2024, 51 (6), pp.e2023GL105143. 10.1029/2023gl105143. hal-04524362

# HAL Id: hal-04524362 https://hal.science/hal-04524362

Submitted on 28 Mar 2024

**HAL** is a multi-disciplinary open access archive for the deposit and dissemination of scientific research documents, whether they are published or not. The documents may come from teaching and research institutions in France or abroad, or from public or private research centers.

L'archive ouverte pluridisciplinaire **HAL**, est destinée au dépôt et à la diffusion de documents scientifiques de niveau recherche, publiés ou non, émanant des établissements d'enseignement et de recherche français ou étrangers, des laboratoires publics ou privés.



Distributed under a Creative Commons Attribution - NonCommercial - NoDerivatives 4.0 International License



# **Geophysical Research Letters**<sup>•</sup>

# **RESEARCH LETTER**

10.1029/2023GL105143

#### **Key Points:**

- High-resolution ensemble and objectoriented approach offer a unique opportunity to study changes in Mediterranean extreme precipitation
- Robust agreement is found for an increase in intensity, volume and severity for future French Mediterranean Heavy Precipitation Events
- Even at convection-permitting scale, considerable uncertainty remains regarding the degree of intensification of the most extreme events

#### **Supporting Information:**

Supporting Information may be found in the online version of this article.

#### Correspondence to:

C. Caillaud, cecile.caillaud@meteo.fr

#### Citation:

Caillaud, C., Somot, S., Douville, H., Alias, A., Bastin, S., Brienen, S., et al. (2024). Northwestern Mediterranean heavy precipitation events in a warmer climate: Robust versus uncertain changes with a large convection-permitting model ensemble. *Geophysical Research Letters*, *51*, e2023GL105143. https://doi.org/10. 1029/2023GL105143

Received 18 JULY 2023 Accepted 19 FEB 2024

© 2024. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NonCommercial-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited, the use is non-commercial and no modifications or adaptations are made.

# Northwestern Mediterranean Heavy Precipitation Events in a Warmer Climate: Robust Versus Uncertain Changes With a Large Convection-Permitting Model Ensemble

Cécile Caillaud<sup>1</sup>, Samuel Somot<sup>1</sup>, Hervé Douville<sup>1</sup>, Antoinette Alias<sup>1</sup>, Sophie Bastin<sup>2</sup>, Susanne Brienen<sup>3</sup>, Marie-Estelle Demory<sup>4,5,6,7</sup>, Andreas Dobler<sup>8</sup>, Hendrik Feldmann<sup>9</sup>, Thomas Frisius<sup>10</sup>, Klaus Goergen<sup>11</sup>, Elizabeth J. Kendon<sup>12</sup>, Klaus Keuler<sup>13</sup>, Geert Lenderink<sup>14</sup>, Paola Mercogliano<sup>15</sup>, Emanuela Pichelli<sup>16</sup>, Pedro M. M. Soares<sup>17</sup>, Merja H. Tölle<sup>18</sup>, and Hylke de Vries<sup>14</sup>

<sup>1</sup>CNRM, Université de Toulouse, Météo-France, CNRS, Toulouse, France, <sup>2</sup>Laboratoire ATmosphere Milieux Observations Spatiales/Institut Pierre Simon Laplace (IPSL), UVSQ Universite Paris-Saclay, Sorbonne Universite, CNRS, CNES, Guyancourt, France, <sup>3</sup>Deutscher Wetterdienst (DWD), Offenbach, Germany, <sup>4</sup>Institute for Atmospheric and Climate Science, ETH Zurich, Zurich, Switzerland, <sup>5</sup>Wyss Academy for Nature, University of Bern, Bern, Switzerland, <sup>6</sup>Climate and Environmental Physics, Physics Institute, University of Bern, Bern, Switzerland, <sup>7</sup>Oeschger Centre for Climate Change Research, University of Bern, Bern, Switzerland, <sup>8</sup>Norwegian Meteorological Institute, Oslo, Norway, <sup>9</sup>Karlsruhe Institute of Technology (KIT), Karlsruhe, Germany, <sup>10</sup>Climate Service Center Germany (GERICS), Helmholtz-Zentrum hereon GmbH, Hamburg, Germany, <sup>11</sup>Institute of Bio- and Geosciences (IBG-3, Agrosphere), Research Centre Jülich, Jülich, Germany, <sup>12</sup>Met Office Hadley Centre, Exeter, UK, <sup>13</sup>Chair of Atmospheric Processes, Brandenburg University of Technology Cottbus-Senftenberg, Cottbus, Germany, <sup>14</sup>Royal Netherlands Meteorological Institute (KNMI), De Bilt, The Netherlands, <sup>15</sup>Fondazione Centro euro-Mediterraneo sui cambiamenti climatici (CMCC), Caserta, Italy, <sup>16</sup>The Abdus Salam International Centre for Theoretical Physics (ICTP), Trieste, Italy, <sup>17</sup>Faculdade de Ciências, Instituto Dom Luiz, Universidade de Lisboa, Lisbon, Portugal, <sup>18</sup>CESR (Center for Environmental Systems Research), University of Kasse, Kassel, Germany

**Abstract** Taking advantage of a large ensemble of Convection Permitting-Regional Climate Models on a pan-Alpine domain and of an object-oriented dedicated analysis, this study aims to investigate future changes in high-impact fall Mediterranean Heavy Precipitation Events at high warming levels. We identify a robust multi-model agreement for an increased frequency from central Italy to the northern Balkans combined with a substantial extension of the affected areas, for a dominant influence of the driving Global Climate Models for projecting changes in the frequency, and for an increase in intensity, area, volume and severity over the French Mediterranean. However, large quantitative uncertainties persist despite the use of convection-permitting models, with no clear agreement in frequency changes over southeastern France and a large range of plausible changes in events' properties, including for the most intense events. Model diversity and international coordination are still needed to provide policy-relevant climate information regarding precipitation extremes.

**Plain Language Summary** Despite growing computational resources and multiple model developments, projecting future changes in the high-impact Mediterranean Heavy Precipitation Events remains both a numerical and scientific challenge. The present study takes advantage of the recent availability of a relatively large ensemble of high resolution Regional Climate Models (2–3 km), which represent a step change in the simulation of precipitation extremes, and of an object-oriented approach, allowing us to track the convective precipitating systems on an hourly basis. Looking at future changes in fall Mediterranean Heavy Precipitation Events at high warming levels, we identify a robust multi-model agreement for an increased frequency from central Italy to the northern Balkans combined with a substantial expansion of the affected areas, and an increase in intensity, area, volume and severity over the French Mediterranean. However, considerable uncertainties remain in terms of frequency over parts of the domain arising from uncertainty in changes in large scale weather patterns, and in terms of degree of intensification for the most intense events. It suggests the need for model diversity and for more coordinated high resolution climate projections with careful selection of different driving global models in order to provide policy-relevant climate information regarding precipitation extremes.

## 1. Introduction

In a context of climate change, future changes in the intensity and frequency of High Impact Weather Events (HIWEs) raise large societal concerns at global and regional levels alike. The northwestern Mediterranean is particularly affected by a typical example of convection-driven HIWEs called Heavy Precipitation Events (HPEs) (Ducrocq et al., 2008; Khodayar et al., 2021; Nuissier et al., 2008, 2011): mainly during the fall season, rainfall amounts greater than 100 mm are recorded in less than a day and often within just a few hours and lead to devastating flash flooding causing many fatalities and economic damage. The occurrence of these extreme events can be explained by favorable meteorological conditions, including synoptic and mesoscale weather features, topography and the proximity of the Mediterranean Sea. The combination of these ingredients leads to the formation of slowly moving or quasi-stationary Mesoscale Convective Systems (MCSs), which explains the exceptional hourly rainfall records.

Concerning observed changes, there is still low confidence in heavy precipitation changes and their attribution over the whole Mediterranean region (Seneviratne et al., 2021). Yet, several studies (Blanchet & Creutin, 2022; Blanchet et al., 2018; Ribes et al., 2019; Vautard et al., 2015) focusing on the French Mediterranean domain showed an observed and potentially human-induced increase in the HPEs intensity and frequency since the middle of the 20th century. The overall trend is consistent with the observed concomitant regional warming and a super Clausius-Clapeyron (CC) scaling of precipitation extremes.

Regarding future changes, the latest IPCC AR6 WP1 report indicates that, at the global scale, extreme daily precipitation events are projected to intensify by about 7% per degree of global warming close to the CC rate (Seneviratne et al., 2021). Yet, when focusing on the Mediterranean region, large uncertainties remain with important discrepancies in the response of Global Climate Models (GCMs, 150 km resolution, e.g. John et al. (2022)) that could be explained by dynamical rather than thermodynamical processes (Pfahl et al., 2017). Multi-model ensembles based on Regional Climate Models (RCMs, 12 km resolution, EURO-CORDEX) still show large uncertainties over this region (Rajczak & Schär, 2017; Tramblay & Somot, 2018; Zittis et al., 2021), with, for example, RX1 day changes ranging from -7 to +17% in Coppola et al. (2021). Part of these regional uncertainties may come from a possible but GCM-dependant decrease in the frequency of favorable weather conditions which could offset thermodynamic increases related to increased humidity as a result of warming and thus make unclear the global intensification of the water cycle.

Moreover, those studies do not provide any information about the response of subdaily precipitation. Yet, short duration rainfall extremes are typical manifestations of Mediterranean HPEs produced through small-scale convective motions. Despite sub-grid parametrizations, state-of-the-art GCMs or RCMs typically used in CMIP or CORDEX are not able to account for this mesoscale dynamical response and the resulting extremes (Ban et al., 2021; Caillaud et al., 2021; Pichelli et al., 2021).

It was only recently that high resolution (typically 2-3 km) Convection-Permitting RCMs (CP-RCMs) could be run in climate mode (Kendon et al., 2021; Lucas-Picher et al., 2021; Prein et al., 2020). This new generation of RCMs allows a step change in the representation of the convective phenomena with a clear added-value compared to lower resolution climate models with parametrized convection to simulate daily and mostly subdaily precipitation extremes. The benefit is particularly evident for Mediterranean HPEs given their typical spatial and time scales (Berthou et al., 2020; Caillaud et al., 2021; Fumière et al., 2020; Pichelli et al., 2021). Moreover, CP-RCMs allow us to go beyond the basic Eulerian approach and to set up an object-oriented Lagrangian tracking to study the propagation and intrinsic characteristics of the extreme convective events. This approach was first applied over North America (Li et al., 2020; Prein et al., 2017, 2020) but has also been used to study European HPEs (Chan et al., 2023; Kahraman et al., 2021), German convective cells (Brisson et al., 2018; Purr et al., 2019, 2021, 2022), as well as Alpine and Northwestern Mediterranean HPEs (Caillaud et al., 2021; Müller et al., 2022, 2023). Thanks to the CORDEX Flagship Pilot Study (FPS) on Convection (Coppola et al., 2020), a first ensemble of kilometer-scale 10-year-timeslice CP-RCM simulations is now available on a common pan-Alpine domain. The extreme precipitation of the hindcast simulations driven by atmospheric reanalyzes has already been evaluated with both Eulerian statistics (Ban et al., 2021) and a Lagrangian approach (Müller et al., 2022). The first Eulerian study focusing on late-century changes (Pichelli et al., 2021), extending the results found by Fumière (2019), foresees a stronger intensification of extreme precipitation at hourly rather than daily timescale, but still with large uncertainties, ranging from about no change to +34% over South-East of France.



In this context, our work aims at providing a more comprehensive and physically oriented study of future changes in the frequency and properties, including intensity, of the Mediterranean HPEs by taking advantage of:

- the recent availability of appropriate modeling and diagnostic tools to study northwestern Mediterranean HPEs under climate change, that is, high resolution CP-RCMs and an object-oriented approach that allows the separation between the frequency and the intrinsic characteristics of convective systems,
- the first-of-its-kind large CP-RCMs ensemble available over the common pan-Alpine domain for strong warming levels.

## 2. Methods

#### 2.1. Models and Simulations

We use a large ensemble of fourteen CP-RCMs simulations performed within the CORDEX FPS on Convection program and protocol, that means high resolution (2–3 km) non-hydrostatic climate simulations, with explicitly resolved deep convection. Six families of CP-RCM models are included in this study, derived from CPM originally developed for National Weather Prediction purposes for AROME, COSMO-CLM or CCLM, UM, and WRF, or newly developed non-hydrostatic high resolution models based on existing RCM (RegCM, REMO).

All the CP-RCM simulations are performed over domains that encompass the common CORDEX FPS pan-Alpine domain called ALP-3 (cf. Figure S1 in Supporting Information S1). Only UM-UKMO covers a large pan-European domain (Berthou et al., 2020).

Due to computational limitations, 10-year only periods are simulated for 3 time slices: 1996–2005, 2041–2050, 2090–2099 (i.e., historical, mid-century and end-century, respectively). The RCP8.5 high-emission scenario (Meinshausen et al., 2011) is chosen for all the future simulations in order to increase the signal-to-noise ratio and highlight projected changes in heavy precipitation despite a background of strong natural variability. Thirteen simulations are available for the mid-century period (all except WRF-FZJ-IDL), as well as for the end-century period (all except REMO-GERICS).

The FPS protocol sets a CP-RCM resolution range, common domain and periods but each institute made its own choice for the simulation setup and the driving RCM and GCM according to their needs and constraints. Most of the simulations are based on a two-tier modeling chain (cf. Figure S1 in Supporting Information S1) with CP-RCMs driven by GCM projections (typically at 150 km resolution) with an intermediate RCM step (12–15 km resolution) to limit the resolution jump.

Further details and references can be found in Table S1 and Figure S2 in Supporting Information S1.

#### 2.2. Mediterranean HPEs Detection and Tracking

To identify and characterize the northwestern Mediterranean heavy precipitating systems in the CP-RCM simulations, we apply a specific detection and tracking algorithm (Caillaud et al., 2021; Morel & Senesi, 2002) onto successive 1-hr accumulated precipitation fields. This method (summarized in Text S1 in Supporting Information S1) has already been used for high-frequency precipitation evaluation in single-model (Caillaud et al., 2021) and multi-model (Müller et al., 2022) studies.

To select French Mediterranean HPEs following Caillaud et al. (2021), a spatial criteria is applied: only the tracks crossing the French Mediterranean area (cf. Figure S1 in Supporting Information S1) are taken into account. Moreover, in addition to the 10 mm/hr intensity threshold set in order to select convective heavy precipitation, only tracks occurring during 24-hr sliding windows with accumulated precipitation exceeding the 100 mm threshold at least over one grid point of the French Mediterranean area are selected. One Mediterranean HPE can thus correspond to a combination of several convective systems (Duffourg et al., 2016), and therefore to several tracks.

All CP-RCMs simulations are firstly interpolated using conservative remapping to the common regular 3 km resolution ALP-3i grid. Then, the tracking algorithm and Mediterranean HPE detection are applied. In order to evaluate the historical simulations, the same methodology is applied to the available kilometric and hourly observations (COMEPHORE over France and GRIPHO over Italy).



The focus is on an extended fall season (from September to December), when most of the HPEs occur in the northwestern Mediterranean region (Ricard et al., 2012). The changes of seasonality of Mediterranean HPEs (studied in Chan et al. (2020)) are not part of this study.

# 3. Results

#### 3.1. Evaluation of the Historical Simulations

The preliminary step before studying future changes is to check the ability of the 14 CP-RCMs historical simulations to reproduce the main characteristics of fall northwestern Mediterranean HPEs, with a focus on the French Mediterranean area with reliable observations. The detailed evaluation can be found in Text S2 in Supporting Information S1.

The main results are the following:

- good spatial representation of the areas known to be affected by fall HPEs in terms of number and positioning, even if there are differences among simulations that cannot be explained by the CP-RCM family or the driving GCM;
- overestimated numbers of HPEs over mountainous regions and Italy that can partly arise from a possible observational rainfall deficit;
- lower numbers of HPEs in the Cevennes foothills with respect to the observations, though to a lesser extent with the AROME simulations. The limitations of CP-RCM in capturing mesoscale convection, such as cold pools (Hirt et al., 2020), may be responsible for this underestimation, which was also found in Müller et al. (2022);
- in the French Mediterranean area, overall good representation of the main characteristics of fall HPEs in terms
  of intensity, area, volume and severity as defined in Text S1 in Supporting Information S1. Duration and
  propagation speed are overestimated by almost all the simulations. For AROME and RegCM simulations
  and even more for REMO, a significant overestimation of rain volumes is found, related to long-lasting and
  oversized systems. RegCM-ICTP simulates too high maximum intensities (Stocchi et al., 2022), whereas
  AROME-CNRM underestimates them (Caillaud et al., 2021).

Therefore, this evaluation allows us to conclude that all our historical timeslice simulations reasonably represent the main present-day characteristics of Mediterranean HPEs, so that it was decided to keep all available GCM/CP-RCM combinations to study their response in a warmer climate.

#### 3.2. Future Changes Under RCP8.5 Scenario

We present here the results for the last decade of the 21st century using an ensemble of 13 CP-RCM simulations. The focus is here mostly on the ensemble-mean response and on the qualitative multi-model agreement (i.e., the number of models that agree about the sign of the anomalies and the statistical significance of sign change at the 90% confidence level using a bootstrapping method described in Text S1 in Supporting Information S1). The results of individual models are presented in Supporting Information S1 (Figures S11–S16).

#### 3.2.1. Changes in Frequency Over the Pan Alpine Domain

Figure 1 presents the ensemble-mean spatial density of precipitating systems above 10 mm/hr and focuses on the fall season over (a) the historical period, (b) the end-century period and (c) the differences between the two periods. An increased frequency is simulated in the major part of the pan-Alpine domain and most strongly in its eastern part, from central Italy to the northern Balkans. If the multi-model ensemble mean does not show any negative anomaly, no change is expected in the south-west and north-east parts of the domain. Moreover, the 10-cell contour isoline of the historical simulation, well representative of the present-day areas known to be affected by Mediterranean HPEs, was superimposed on the projected anomalies and thus highlights a substantial extension of the affected areas. At the end of 21st century, the pre-Alps, the interior of Italy, Slovenia or Croatia, the south of Austria or Switzerland may be affected by intense convective systems during the fall, whereas they are hardly impacted nowadays. The land area with densities above 10 cells per year is expected to almost double by the end of the century, from 214002 to 411,867 km<sup>2</sup>, that is, +92%.





Figure 1. Spatial density of cells above 10 mm/hr during SOND for the multi-model mean for the historical (a) and the end-century (b) periods. Raw differences (c) with difference of number of cells per year over land indicated at the top right and in orange, the 10-cell contour isolines for the historical period. Model agreement (d) for the sign changes in %. On all figures, orography (750 m) plotted in brown lines.

Figure 1d highlights the good agreement among the different simulations. About 95%, 83% and 71% of the land domain exhibit a model agreement above 67%, 80%, and 90% respectively. However, there is no clear model agreement over the Southeast of France, the Northwest of Italy and a large portion of the Mediterranean sea. Looking at individual model changes (cf. Figure S11 in Supporting Information S1), the large increase in the frequency of heavy precipitation is also found over central Italy to the northern Balkans for all simulations. If we classify the simulations according to the driving GCM, we find some common behaviors: for example, the three simulations driven by EC-Earth r12 present lower changes, with differences of annual number of cells above 10 mm/hr over land between +1.7 and +1.9 to be compared to the +3.4 of the ensemble mean. In contrast, no common behavior can be easily identified among the CP-RCM families. Such results highlight a dominant influence of the driving GCM for predicting changes in the number of HPEs, pointing the dominant role of the change in large scale dynamics and natural variability in setting the frequency change of the Mediterranean HPEs.

The lack of consensus on frequency changes over parts of the ALP-3 domain leads us to investigate whether consensus can be found for the other characteristics of Mediterranean HPEs. The French Mediterranean area, where a reliable evaluation of the simulations has been carried out, provides an appropriate study area.

### 3.2.2. Focus on the French Mediterranean

Figure 2a shows the multi-model ensemble changes of French Mediterranean HPEs' characteristics projected at the end of the century and spatially averaged over the area. For the change in the number of tracks ("Number"), the lack of consensus is confirmed: the ensemble-mean increase in the number of tracks (+19%) hides indeed a large range of anomalies across the individual simulations between -12% and +125%. Moreover, a majority of simulations shows a limited and no significant sign change (between -12% and +8%). Three simulations anticipate larger and significant increases of which the two driven by HadGEM2-ES and the UM-UKMO simulation driven by HadGEM3-GC with an outstanding increase of +125%. These changes can be linked with the changes in the number of days exceeding 100 mm/d ("Nb100"), a proxy of the number of days with favorable large-scale conditions for HPEs. This highlights again a dominant influence of the driving GCM for predicting changes in the number of HPEs.

A full model agreement, which is also highly statistically significant, is found for an increase in mean and maximum precipitation intensities, ranging from +0% to +11% and from +3% to +20%, respectively. For changes in mean and maximum area, there is more inter-model spread in the magnitude, but a good agreement (12/13 simulations) for an increase (ranging from -23% to +28% and from -24% to +20% respectively). In line with the projected increase in intensities and areas, 12 out of 13 simulations show an increase in volume and severity (with a large range of values among simulations, from -9% to +87% and from -2% to +56%, respectively). A good model agreement is also found for little and no significant change in duration and in propagation speed. Future increases in areas are also found in Müller et al. (2023) for all-year Mediterranean HPEs, in Purr et al. (2021) for convective cells over Germany and in Prein et al. (2017) for intense summertime MCSs over North-America.

Yet, the UM-UKMO behaves differently for most characteristics: in addition to the highest number of tracks, the simulation shows a larger decrease in duration and is the only one to propose a decrease in areas (around -25%). On the other hand, it presents the highest increase in mean and maximum intensity (+11% and +20% respectively). These contrasting and significant changes explain non-significant changes in volume and severity. This simulation thus projects different changes in convection over the French Mediterranean, in line with the changes in European storms found by Chan et al. (2023) comparing the same UM-UKMO simulation to a CCLM-ETHZ simulation over a pan-European domain, with the latter using a pseudo-global warming approach: more frequent, smaller storms with UM-UKMO contrast with fewer, larger ones with CCLM-ETHZ. In this case, differences in changes in the large scale dynamics are identified as being responsible for the differences in changes in storm characteristics.

In order to combine different tracks characteristics, we also present IDH plots (maximum Intensity Duration Histogram, cf. Figures 2b–2d) and IAH plots (maximum Intensity maximum Area Histogram, cf. Figures 2e–2g) which present the distribution of tracks' frequency defined as the combined occurrence of maximum intensity and duration or area values in chosen bins, normalized by the total number of tracks. The IDH and IAH plots for the multi-model frequency ensemble mean confirm the increase of intensity expected at the end of the century for HPEs. For intensities below 30 mm/hr, the frequency of short and small events increases. Above 30 mm/hr, an increase of the frequencies of heavy precipitating systems whatever the duration and the area is expected. The individual changes in IDH distributions (cf. Figure S12 in Supporting Information S1) are rather patchy. For the simulations driven by MPI-ESM-LR or EC-Earth r14, it is difficult to find a clear signal. For the others, the pattern is close to the multi-model mean's one, in particular those driven by HadGEM2-ES and for the UM-UKMO simulation, which shows the clearest increase in frequency of short duration events. The panel for IAH changes (cf. Figure S13 in Supporting Information S1) displays a little more dispersion, but still with a clearer signal of increased frequency of small-sized events for UM-UKMO.

Understanding these differences deserves further investigation which is however beyond the scope of the present study. Multiple differences in the model set-up may contribute to such differences and suggest the need for even more coordinated and comprehensive intercomparisons or additional sensitivity tests to allow a better interpretation of the results.

The full agreement for an increase in intensity of the future fall French Mediterranean convective systems is however noteworthy when compared to other past studies such as Tramblay and Somot (2018) on French Mediterranean catchment basins at daily scale, but the values are relatively low, ranging from +3% to +20%. Normalizing these values by the corresponding fall (SOND) changes in mean 2 m temperature averaged over the





Figure 2.

19448007, 2024, 6, Downloaded from https://agupubs.onlinelibrary.wiley.com/doi/10.1029/2023GL105143 by Cochrane France, Wiley Online Library on [28/03/2024], See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL105143 by Cochrane France, Wiley Online Library on [28/03/2024], See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL105143 by Cochrane France, Wiley Online Library on [28/03/2024], See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL105143 by Cochrane France, Wiley Online Library on [28/03/2024], See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL105143 by Cochrane France, Wiley Online Library on [28/03/2024], See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL105143 by Cochrane France, Wiley Online Library on [28/03/2024], See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL105143 by Cochrane France, Wiley Online Library on [28/03/2024], See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL105143 by Cochrane France, Wiley Online Library on [28/03/2024], See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL105143 by Cochrane France, Wiley Online Library on [28/03/2024], See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL105143 by Cochrane France, Wiley Online Library on [28/03/2024], See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL105143 by Cochrane France, Wiley Online Library on [28/03/2024], See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL105143 by Cochrane France, Wiley Online Library on [28/03/2024], See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2023GL105143 by Cochrane France, Wiley Online Library on [28/03/2024], See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/10.1029/2024], See the Terms and Conditions (https://onlinelibrary.wiley.com/doi/1

whole ALP-3 domain (see Table S2 in Supporting Information S1) leads to a range from +1 to +5%/°C, below the expected +7%/°C CC rate (Vergara-Temprado et al., 2021) and therefore well below the super-CC rate that has been reported for extreme hourly precipitation in previous studies (Kendon et al., 2014; Lenderink et al., 2011, 2017, 2019). Although our study is not aligned with these previous studies, a stronger intensification could be found for the most extreme precipitation events (Purr et al., 2021), which leads us to look at the tail distribution of the events.

Therefore, we focus now on the most intense tracks, that is, those with maximum intensities above each simulation's historical 97th percentile of maximum intensities. This fixed value is chosen to be consistent in the observations with the thresholds used in the operational weather warning system of Météo-France for heavy precipitation in southeastern France, but also to have a sufficient number of convective systems.

Changes in the main characteristics of the most intense tracks (cf. Figure S14 in Supporting Information S1) show now a full model agreement for a strong increase in frequency, ranging from +38% to +326%, still large model agreement for the increase in intensity but larger spread for changes in spatial extent, and hence volume and severity of HPEs. The individual distributions of the maximum intensities (cf. Figure S15 in Supporting Information S1) illustrate the difference of behavior among simulations. Most of the historical simulations underestimate the observed highest values but reach very different maximum values. The differences are even larger for end-century simulations, with, for example, much higher values for UM-UKMO or WRF-IPSL. RegCM-ICTP behaves differently with very high values, some of them exceeding 300 m/hr, in both historical and endcentury simulations.

Despite the limited length of the simulations, our tracking methodology allows to identify a large number of tracks (at least 155/year) and therefore to use a Generalized Pareto Distribution (GPD, presented in Text S1 in Supporting Information S1) for modeling the tails of the distribution of the tracks' maximum intensities. In particular, this method allows us to infer intensity changes of events with return periods exceeding 10 years. Individual return period plots are presented in Figure S16 in Supporting Information S1. Figure 2h focuses on the 20-year return values and mean relative changes between historical and end-century periods for each simulation, except RegCM-ICTP which is the only one to project a decrease associated with much higher values and very large confidence intervals (see Table S2 in Supporting Information S1 for details). Most historical simulations show lower than observed values. For the end-century period, increases in 20-year return values are higher than when looking at changes in intensity for all tracks, still with a large range among simulations, from +16% to +95%. Based on the value calculated from observations, this range of increase would correspond to 20-year return values between 142 and 239 mm/hr at the end of the century. These very high values must be set against the record values already measured: world record of 305 mm/hr (June 1947, Missouri, USA), southeastern France record of 160 mm/hr (October 1986, Pyrénées-Orientales). When compared to the average temperature increase (see Table S2 in Supporting Information S1), we also find very large differences, with values (+4 to +27%/°C) ranging from below the CC rate up to 4 times the CC rate. Again, it is not possible to find common behavior within CP-RCM family or according to the driving GCM. For instance, two CP-RCMs of the same family driven by the same GCM show the smallest and largest increase respectively. Likewise, there is no direct relationship between the increase in intensity for 20-year return periods and the increase in frequency and neither when comparing to the increase in regional near-surface temperature, highlighting that the HPE future evolution can deviate substantially from the assessed global picture of an increase at the CC rate (Seneviratne et al., 2021).

For the mid-century period (see Text S3 in Supporting Information S1 for details), the conclusions are close to those given for the end-of-century period; although with lower values and less statistical significance for the changes of the main characteristics, and a stronger impact of the climate natural variability on the frequency change.

**Figure 2.** For the selection of tracks over the French Mediterranean area during SOND: main characteristics' differences (a) between the end-century and the historical periods for the multi-model mean and the 13 CP-RCMs (one color per family of CP-RCM and one symbol per driving GCM). Model agreement on the sign change in color if higher than 80%, otherwise in black and the number of simulations with significant sign changes in brackets. Multi-model mean for IDH and IAH plots for the historical (b, e) and the end-century (c, f) periods, and the differences (d, g). Individual values of maximum intensity (h) for 20-year return periods given by a GPD fit for the most intense tracks (except RegCM) with the same colors and symbols as (a).



## 4. Conclusions

Taking advantage of the unprecedented large ensemble of Convection Permitting-Regional Climate Models (CP-RCMs) deployed in the CORDEX Flagship Pilot Study (FPS) on Convection on a common pan-Alpine domain and combining it with an object-oriented approach, this study aims to investigate the future changes in fall Mediterranean Heavy Precipitation Events (HPEs) at high warming levels. A convective-cell detection and tracking algorithm is applied to 1-hr accumulated precipitation fields of fourteen CP-RCM simulations: the 10-year historical simulations are first evaluated with regards to the observations, then compared to mid-century and end-century timeslices to assess future changes. This Lagrangian approach is used to analyze both the frequency and the intrinsic characteristics of either all or the most extreme convective systems. The multi-model approach allows us to quantify model uncertainties, assess the qualitative agreement among the available simulations, and thus identify the most robust changes.

The following robust results were identified:

- All CP-RCM historical simulations reasonably represent the main present-day characteristics of Mediterranean HPEs.
- At high warming levels, an increased frequency is simulated in the greater part of the pan-Alpine domain and most strongly in its eastern part where there is full agreement among simulations. This increase in frequency is combined with a substantial extension, that is, a doubling of the affected areas.
- The study highlights a dominant influence of the driving GCM for projecting changes in the frequency of HPEs. Hence, the choice of the driving GCM is essential and should be done through studies on GCM simulations focusing on favorable weather regimes such as Nuissier et al. (2011) or Brigode et al. (2018) using a storyline perspective (Shepherd, 2019; Zappa & Shepherd, 2017).
- Over the French Mediterranean, a full model agreement is found for an increase in mean and maximum intensities (ranging from +0% to +11% and from +3% to +20% respectively) for all tracks. Large agreement is also outlined for an increase in mean and maximum areas and therefore in volume and severity. A relatively good agreement is also found for little change in duration and propagation speed.

However, large quantitative uncertainties persist despite the use of CP-RCMs:

- Even with high-resolution modeling and explicitly resolved deep convection, significant biases in present-day precipitation extremes still remain, as also highlighted in the DYAMOND global CPM simulations (Feng et al., 2023).
- No clear model agreement is found in frequency changes over part of the pan-Alpine domain including southeastern France, probably linked to differences in large-scale synoptic conditions imposed by the driving GCMs.
- A large range of plausible changes in HPE properties is simulated across the ensemble of GCM/CP-RCM combinations, including for the scaling of increased intensity with the corresponding regional warming.
- One simulation, with a distinctive set-up, projects different, yet plausible, changes in convection with more intermittent but more intense convective systems with shorter lifetime and smaller size.
- Regarding the most intense HPEs over the French Mediterranean, the large range of projected changes in frequency and in 20-year return maximum intensity values (up to +95%) cannot be easily related to the choice of the CP-RCM or driving GCM, or to the projected regional warming.

Various limitations of the present study provide some lessons for future studies:

- 10-year simulations are not long enough to assess accurately changes in extremes such as Mediterranean HPEs
  which show a high interannual variability, as confirmed by the mid-century study. Similarly, simulations over
  larger domains would allow CP-RCMs to express their own internal variability and their intrinsic sensitivity,
  potentially leading to a different signal to noise ratio.
- 2 to 3-km convection-permitting models allow us to switch off the parametrization of deep convection, but still
  need some parametrization such as turbulence or micro-physical processes, which may also contribute to
  model uncertainties and also need specific assessments. Therefore, increasing model resolution will not
  necessarily lead to a reduction of uncertainty and should not lead to decreasing model diversity and/or the size
  of multi-model ensembles.
- Although the CORDEX FPS on Convection has made possible this first CP-RCM intercomparison, it remains
  an ensemble of opportunity with limited coordination regarding the driving method and GCM strategy. Larger

and better designed ensembles are thus needed, possibly combined with a new hybrid downscaling technique such as a RCM emulator (Doury et al., 2022).

 Additional impact variables (hail, wind gusts) or process-oriented diagnostics (composites of environmental variables such as the vertical profile of temperature and humidity) are needed to better document and understand future changes in Mediterranean HPEs.

Our final message is thus that efforts to provide policy-relevant climate information regarding future changes in precipitation extremes using ensembles of high-resolution regional models should be continued and will need enhanced coordination, at least at the European level as a follow-up to the CORDEX FPS on Convection.

## **Data Availability Statement**

The hourly precipitation data of the CORDEX-FPS on Convection CP-RCMs ensemble are or will soon become available on the ESGF archive. COMEPHORE data set is available through AERIS platform (https://radarsmf. aeris-data.fr/en/home-page/). The archiving of the tracking data over the whole ALP-3 domain and the selection of tracks over the French Mediterranean for all simulations (historical, mid-century and end-century periods) and observation is available on Zenodo (Caillaud, 2023). Codes for data visualization are based on the NCL language (NCAR, 2019) and the Python library (Hunter, 2007).

#### References

- Ban, N., Caillaud, C., Coppola, E., Pichelli, E., Sobolowski, S., Adinolfi, M., et al. (2021). The first multi-model ensemble of regional climate simulations at kilometer-scale resolution, part I: Evaluation of precipitation. *Climate Dynamics*, 57(1–2), 275–302. https://doi.org/10.1007/ s00382-021-05708-w
- Berthou, S., Kendon, E. J., Chan, S. C., Ban, N., Leutwyler, D., Schär, C., & Fosser, G. (2020). Pan-European climate at convection-permitting scale: A model intercomparison study. *Climate Dynamics*, 55(1–2), 35–59. https://doi.org/10.1007/s00382-018-4114-6
- Blanchet, J., & Creutin, J.-D. (2022). Instrumental agreement and retrospective analysis of trends in precipitation extremes in the French Mediterranean Region. *Environmental Research Letters*, 17(7), 074011. https://doi.org/10.1088/1748-9326/ac7734
- Blanchet, J., Molinié, G., & Touati, J. (2018). Spatial analysis of trend in extreme daily rainfall in southern France. *Climate Dynamics*, 51(3), 799–812. https://doi.org/10.1007/s00382-016-3122-7
- Brigode, P., Gérardin, M., Bernardara, P., Gailhard, J., & Ribstein, P. (2018). Changes in French weather pattern seasonal frequencies projected by a CMIP5 ensemble. *International Journal of Climatology*, 38(10), 3991–4006. https://doi.org/10.1002/joc.5549
- Brisson, E., Brendel, C., Herzog, S., & Ahrens, B. (2018). Lagrangian evaluation of convective shower characteristics in a convection-permitting model. *Meteorologische Zeitschrift*, 27(1), 59–66. https://doi.org/10.1127/metz/2017/0817
- Caillaud, C. (2023). Mediterranean heavy precipitation events in a warmer climate: Robust versus uncertain changes with a large convectionpermitting model ensemble: Supporting information (version v20240112) [Dataset]. Zenodo. https://doi.org/10.5281/zenodo.8124196
- Caillaud, C., Somot, S., Alias, A., Bernard-Bouissières, I., Fumière, Q., Laurantin, O., et al. (2021). Modelling mediterranean heavy precipitation events at climate scale: An object-oriented evaluation of the CNRM-AROME convection-permitting regional climate model. *Climate Dy*namics, 56(5–6), 1–36. https://doi.org/10.1007/s00382-020-05558-y
- Chan, S. C., Kendon, E. J., Berthou, S., Fosser, G., Lewis, E., & Fowler, H. J. (2020). Europe-wide precipitation projections at convection permitting scale with the Unified Model. *Climate Dynamics*, 55(3–4), 409–428. https://doi.org/10.1007/s00382-020-05192-8
- Chan, S. C., Kendon, E. J., Fowler, H. J., Kahraman, A., Crook, J., Ban, N., & Prein, A. F. (2023). Large-scale dynamics moderate impact-relevant changes to organised convective storms. *Communications Earth & Environment*, 4(1), 8. https://doi.org/10.1038/s43247-022-00669-2
- Coppola, E., Nogherotto, R., Ciarlo, J. M., Giorgi, F., van Meijgaard, E., Kadygrov, N., et al. (2021). Assessment of the European climate projections as simulated by the large EURO-CORDEX regional and global climate model ensemble. *Journal of Geophysical Research: At*mospheres, 126(4). https://doi.org/10.1029/2019JD032344
- Coppola, E., Sobolowski, S., Pichelli, E., Raffaele, F., Ahrens, B., Anders, I., et al. (2020). A first-of-its-kind multi-model convection permitting ensemble for investigating convective phenomena over Europe and the Mediterranean. *Climate Dynamics*, 55(1–2), 3–34. https://doi.org/10. 1007/s00382-018-4521-8
- Doury, A., Somot, S., Gadat, S., Ribes, A., & Corre, L. (2022). Regional climate model emulator based on deep learning: Concept and first evaluation of a novel hybrid downscaling approach. *Climate Dynamics*, 1–29(5–6), 1751–1779. https://doi.org/10.1007/s00382-022-06343-9
- Ducrocq, V., Nuissier, O., Ricard, D., Lebeaupin, C., & Thouvenin, T. (2008). A numerical study of three catastrophic precipitating events over southern France. II: Mesoscale triggering and stationarity factors. *Quarterly Journal of the Royal Meteorological Society*, 134(630), 131–145. https://doi.org/10.1002/qj.199
- Duffourg, F., Nuissier, O., Ducrocq, V., Flamant, C., Chazette, P., Delanoë, J., et al. (2016). Offshore deep convection initiation and maintenance during the HyMeX IOP 16a heavy precipitation event. *Quarterly Journal of the Royal Meteorological Society*, 142(S1), 259–274. https://doi. org/10.1002/qj.2725
- Feng, Z., Leung, L. R., Hardin, J., Terai, C. R., Song, F., & Caldwell, P. (2023). Mesoscale convective systems in DYAMOND global convectionpermitting simulations. *Geophysical Research Letters*, 50(4), e2022GL102603. https://doi.org/10.1029/2022GL102603
- Fumière, Q. (2019). Changement climatique et précipitations extrêmes: Apport des modèles résolvant la convection (Unpublished doctoral dissertation). Université Paul Sabatier, Toulouse III. Retrieved from http://thesesups.ups-tlse.fr/4564/
- Fumière, Q., Déqué, M., Nuissier, O., Somot, S., Alias, A., Caillaud, C., et al. (2020). Extreme rainfall in mediterranean France during the fall: Added value of the CNRM-AROME convection-permitting regional climate model. *Climate Dynamics*, 55(1–2), 77–91. https://doi.org/10. 1007/s00382-019-04898-8
- Hirt, M., Craig, G. C., Schäfer, S. A., Savre, J., & Heinze, R. (2020). Cold-pool-driven convective initiation: Using causal graph analysis to determine what convection-permitting models are missing. *Quarterly Journal of the Royal Meteorological Society*, 146(730), 2205–2227. https://doi.org/10.1002/qj.3788

#### Acknowledgments

We acknowledge WCRP-CORDEX-FPS on Convective phenomena at high resolution over Europe and the Mediterranean (FPSCONV-ALP-3). This work was funded by European Union through Horizon-2020 EUCP (Grant 776613) and HORIZON-EUROPE IMPETUS4CHANGE projects and by ANR-France-2030 as part of PEPR TRACCS programme (LOCALISING, ANR-22-EXTR-0011). We thank the research data exchange infrastructure and services provided by Jülich Supercomputing Centre, Germany, as part of Helmholtz Data Federation initiative. CETEMPS, University of L'Aquila, for allowing ICTP to access the Italian database of precipitation which GRIPHO is based on, AERIS for providing COMEPHORE data set and Aurelien Ribes for his statistical advice. HdV thanks Bert van Ulft, Erik van Meiigaard for KNMI simulations. KGo acknowledges the computing time on supercomputer JURECA at Forschungszentrum Jülich (Grant CJJSC39). MED acknowledges partnership for advanced computing in Europe for awarding her access to Piz Daint at the Swiss National Supercomputing Centre, as well as the Federal Office for Meteorology and Climatology, Center for Climate Systems Modeling and ETH Zurich for their contributions to COSMO-crCLIM development. IPSL's work was granted access to HPC resources of TGCC under allocations 2019-A0030106877 and 2020-A0030106877 made by GENCI, and to IPSL's ESPRI computing/storage resources

- Hunter, J. D. (2007). Matplotlib: A 2D graphics environment [Software]. Computing in Science & Engineering, IEEE COMPUTER SOC, 9(3), 90–95. https://doi.org/10.1109/MCSE.2007.55
- John, A., Douville, H., Ribes, A., & Yiou, P. (2022). Quantifying CMIP6 model uncertainties in extreme precipitation projections. Weather and Climate Extremes, 36, 100435. https://doi.org/10.1016/j.wace.2022.100435
- Kahraman, A., Kendon, E. J., Chan, S. C., & Fowler, H. J. (2021). Quasi-stationary intense rainstorms spread across Europe under climate change. *Geophysical Research Letters*, 48(13), e2020GL092361. https://doi.org/10.1029/2020GL092361
- Kendon, E. J., Prein, A. F., Senior, C., & Stirling, A. (2021). Challenges and outlook for convection-permitting climate modelling. *Philosophical Transactions of the Royal Society A*, 379(2195), 20190547. https://doi.org/10.1098/rsta.2019.0547
- Kendon, E. J., Roberts, N. M., Fowler, H. J., Roberts, M. J., Chan, S. C., & Senior, C. A. (2014). Heavier summer downpours with climate change revealed by weather forecast resolution model. *Nature Climate Change*, 4(7), 570–576. https://doi.org/10.1038/nclimate2258
- Khodayar, S., Davolio, S., Di Girolamo, P., Lebeaupin Brossier, C., Flaounas, E., Fourrie, N., et al. (2021). Overview towards improved understanding of the mechanisms leading to heavy precipitation in the western mediterranean: Lessons learned from HyMeX. Atmospheric Chemistry and Physics, 21(22), 17051–17078. https://doi.org/10.5194/acp-21-17051-2021
- Lenderink, G., Barbero, R., Loriaux, J., & Fowler, H. (2017). Super-Clausius–Clapeyron scaling of extreme hourly convective precipitation and its relation to large-scale atmospheric conditions. *Journal of Climate*, 30(15), 6037–6052. https://doi.org/10.1175/JCLI-D-16-0808.1
- Lenderink, G., Belušić, D., Fowler, H. J., Kjellström, E., Lind, P., van Meijgaard, E., et al. (2019). Systematic increases in the thermodynamic response of hourly precipitation extremes in an idealized warming experiment with a convection-permitting climate model. *Environmental Research Letters*, 14(7), 074012. https://doi.org/10.1088/1748-9326/ab214a
- Lenderink, G., Mok, H., Lee, T., & Van Oldenborgh, G. (2011). Scaling and trends of hourly precipitation extremes in two different climate zones–Hong Kong and the Netherlands. *Hydrology and Earth System Sciences*, 15(9), 3033–3041. https://doi.org/10.5194/hess-15-3033-2011
- Li, L., Li, Y., & Li, Z. (2020). Object-based tracking of precipitation systems in western Canada: The importance of temporal resolution of source data. *Climate Dynamics*, 1–17(9–10), 2421–2437. https://doi.org/10.1007/s00382-020-05388-y
- Lucas-Picher, P., Argüeso, D., Brisson, E., Tramblay, Y., Berg, P., Lemonsu, A., et al. (2021). Convection-permitting modeling with regional climate models: Latest developments and next steps. Wiley Interdisciplinary Reviews: Climate Change, 12(6), e731. https://doi.org/10.1002/ wcc.731
- Meinshausen, M., Smith, S. J., Calvin, K., Daniel, J. S., Kainuma, M. L., Lamarque, J.-F., et al. (2011). The RCP greenhouse gas concentrations and their extensions from 1765 to 2300. *Climatic Change*, 109(1–2), 213–241. https://doi.org/10.1007/s10584-011-0156-z
- Morel, C., & Senesi, S. (2002). A climatology of mesoscale convective systems over Europe using satellite infrared imagery. I: Methodology. Quarterly Journal of the Royal Meteorological Society: A journal of the atmospheric sciences, applied meteorology and physical oceanography, 128(584), 1953–1971. https://doi.org/10.1256/003590002320603485
- Müller, S. K., Caillaud, C., Chan, S., de Vries, H., Bastin, S., Berthou, S., et al. (2022). Evaluation of Alpine-Mediterranean precipitation events in convection-permitting regional climate models using a set of tracking algorithms. *Climate Dynamics*, 1–19(1–2), 939–957. https://doi.org/10. 1007/s00382-022-06555-z
- Müller, S. K., Pichelli, E., Coppola, E., Berthou, S., Brienen, S., Caillaud, C., et al. (2023). The climate change response of heavy precipitation events over the Alps and in the mediterranean. *Climate Dynamics*, 62(1), 165–186. https://doi.org/10.1007/s00382-023-06901-9

NCAR. (2019). The NCAR command language (version 6.6.2) [Software]. UCAR/NCAR/CISL/TDD. https://doi.org/10.5065/D6WD3XH5 Nuissier, O., Ducrocq, V., Ricard, D., Lebeaupin, C., & Anquetin, S. (2008). A numerical study of three catastrophic precipitating events over

Southern France. I: Numerical framework and synoptic ingredients. Quarterly Journal of the Royal Meteorological Society, 134(630), 111–130. https://doi.org/10.1002/qj.200

- Nuissier, O., Joly, B., Joly, A., Ducrocq, V., & Arbogast, P. (2011). A statistical downscaling to identify the large-scale circulation patterns associated with heavy precipitation events over southern France. *Quarterly Journal of the Royal Meteorological Society*, 137(660), 1812–1827. https://doi.org/10.1002/qj.866
- Pfahl, S., O'Gorman, P. A., & Fischer, E. M. (2017). Understanding the regional pattern of projected future changes in extreme precipitation. *Nature Climate Change*, 7(6), 423–427. https://doi.org/10.1038/nclimate3287
- Pichelli, E., Coppola, E., Sobolowski, S., Ban, N., Giorgi, F., Stocchi, P., et al. (2021). The first multi-model ensemble of regional climate simulations at kilometer-scale resolution part 2: Future precipitation projections. *Climate Dynamics*, 56(11–12), 3581–3602. https://doi.org/10. 1007/s00382-021-05657-4
- Prein, A. F., Liu, C., Ikeda, K., Bullock, R., Rasmussen, R. M., Holland, G. J., & Clark, M. (2020). Simulating North American mesoscale convective systems with a convection-permitting climate model. *Climate Dynamics*, 55(1–2), 95–110. https://doi.org/10.1007/s00382-017-3993-2
- Prein, A. F., Liu, C., Ikeda, K., Trier, S. B., Rasmussen, R. M., Holland, G. J., & Clark, M. P. (2017). Increased rainfall volume from future convective storms in the US. *Nature Climate Change*, 7(12), 880–884. https://doi.org/10.1038/s41558-017-0007-7

Purr, C., Brisson, E., & Ahrens, B. (2019). Convective shower characteristics simulated with the convection-permitting climate model COSMO– CLM. Atmosphere, 10(12), 810. https://doi.org/10.3390/atmos10120810

- Purr, C., Brisson, E., & Ahrens, B. (2021). Convective rain cell characteristics and scaling in climate projections for Germany. International Journal of Climatology, 41(5), 3174–3185. https://doi.org/10.1002/joc.7012
- Purr, C., Brisson, E., Schlünzen, K. H., & Ahrens, B. (2022). Convective rain cell properties and the resulting precipitation scaling in a warmtemperate climate. *Quarterly Journal of the Royal Meteorological Society*, 148(745), 1768–1781. https://doi.org/10.1002/qj.4277
- Rajczak, J., & Schär, C. (2017). Projections of future precipitation extremes over Europe: A multimodel assessment of climate simulations. Journal of Geophysical Research: Atmospheres, 122(20), 10–773. https://doi.org/10.1002/2017JD027176
- Ribes, A., Thao, S., Vautard, R., Dubuisson, B., Somot, S., Colin, J., et al. (2019). Observed increase in extreme daily rainfall in the French Mediterranean. *Climate Dynamics*, 1–20(1–2), 1095–1114. https://doi.org/10.1007/s00382-018-4179-2
- Ricard, D., Ducrocq, V., & Auger, L. (2012). A climatology of the mesoscale environment associated with heavily precipitating events over a northwestern Mediterranean area. *Journal of Applied Meteorology and Climatology*, 51(3), 468–488. https://doi.org/10.1175/JAMC-D-11-017.1
- Seneviratne, S. I., Zhang, X., Adnan, M., Badi, W., Dereczynski, C., Di Luca, A., et al. (2021). Chapter 11: Weather and climate extreme events in a changing climate. In *Ipcc 2021: Climate Change 2021: The physical science basis. Contribution of Working Group 1 to the Sixth Assessment Report of the Intergovernmental Panel on Climate Change*. https://doi.org/10.1017/9781009157896.013
- Shepherd, T. G. (2019). Storyline approach to the construction of regional climate change information. *Proceedings of the Royal Society A*, 475(2225), 20190013. https://doi.org/10.1098/rspa.2019.0013

- Stocchi, P., Pichelli, E., Torres Alavez, J. A., Coppola, E., Giuliani, G., & Giorgi, F. (2022). Non-hydrostatic RegCM4 (RegCM4-NH): Evaluation of precipitation statistics at the convection-permitting scale over different domains. *Atmosphere*, 13(6), 861. https://doi.org/10.3390/ atmos13060861
- Tramblay, Y., & Somot, S. (2018). Future evolution of extreme precipitation in the Mediterranean. *Climatic Change*, 151(2), 289–302. https://doi.org/10.1007/s10584-018-2300-5
- Vautard, R., Yiou, P., van Oldenborgh, G.-J., Lenderink, G., Thao, S., Ribes, A., et al. (2015). Extreme fall 2014 precipitation in the Cévennes mountains. Bulletin of the American Meteorological Society, 96(12), S56–S60. https://doi.org/10.1175/bams-eee\_2014\_ch12.1
- Vergara-Temprado, J., Ban, N., & Schär, C. (2021). Extreme sub-hourly precipitation intensities scale close to the Clausius-Clapeyron rate over Europe. *Geophysical Research Letters*, 48(3), e2020GL089506. https://doi.org/10.1029/2020GL089506
- Zappa, G., & Shepherd, T. G. (2017). Storylines of atmospheric circulation change for European regional climate impact assessment. Journal of Climate, 30(16), 6561–6577. https://doi.org/10.1175/JCLI-D-16-0807.1
- Zittis, G., Bruggeman, A., & Lelieveld, J. (2021). Revisiting future extreme precipitation trends in the Mediterranean. Weather and Climate Extremes, 34, 100380. https://doi.org/10.1016/j.wace.2021.100380

# **References From the Supporting Information**

- Adinolfi, M., Raffa, M., Reder, A., & Mercogliano, P. (2021). Evaluation and expected changes of summer precipitation at convection permitting scale with COSMO-CLM over alpine space. *Atmosphere*, *12*(1), 54. https://doi.org/10.3390/atmos12010054
- Baldauf, M., Seifert, A., Förstner, J., Majewski, D., Raschendorfer, M., & Reinhardt, T. (2011). Operational convective-scale numerical weather prediction with the COSMO model: Description and sensitivities. *Monthly Weather Review*, 139(12), 3887–3905. https://doi.org/10.1175/ MWR-D-10-05013.1
- Belušic, D., de Vries, H., Dobler, A., Landgren, O., Lind, P., Lindstedt, D., et al. (2020). HCLIM38: A flexible regional climate model applicable for different climate zones from coarse to convection permitting scales. *Geoscientific Model Development*, 13(3), 1311–1333. https://doi.org/ 10.5194/gmd-13-1311-2020
- Brousseau, P., Seity, Y., Ricard, D., & Léger, J. (2016). Improvement of the forecast of convective activity from the AROME-France system. *Quarterly Journal of the Royal Meteorological Society*, 142(699), 2231–2243. https://doi.org/10.1002/qj.2822
- Coles, S., Bawa, J., Trenner, L., & Dorazio, P. (2001). In An introduction to statistical modeling of extreme values (Vol. 208). Springer. https://doi.org/10.1007/978-1-4471-3675-0
- de Vries, H., Lenderink, G., van der Wiel, K., & van Meijgaard, E. (2022). Quantifying the role of the large-scale circulation on European summer precipitation change. *Climate Dynamics*, 59(9–10), 2871–2886. https://doi.org/10.1007/s00382-022-06250-z
- Dufresne, J.-L., Foujols, M.-A., Denvil, S., Caubel, A., Marti, O., Aumont, O., et al. (2013). Climate change projections using the IPSL-CM5 Earth system model: From CMIP3 to CMIP5. *Climate Dynamics*, 40(9–10), 2123–2165. https://doi.org/10.1007/s00382-012-1636-1
- Fantini, A. (2019). Climate change impact on flood hazard over Italy (Unpublished doctoral dissertation). Universita degli Studi di Trieste. Retrieved from http://hdl.hendle.net/11368/2940009
- Giorgi, F., Coppola, E., Solmon, F., Mariotti, L., Sylla, M., Bi, X., et al. (2012). RegCM4: Model description and preliminary tests over multiple CORDEX domains. *Climate Research*, 52, 7–29. https://doi.org/10.3354/cr01018
- Hazeleger, W., Wang, X., Severijns, C., Ştefănescu, S., Bintanja, R., Sterl, A., et al. (2012). EC-Earth V2. 2: Description and validation of a new seamless earth system prediction model. *Climate Dynamics*, 39(11), 2611–2629. https://doi.org/10.1007/s00382-011-1228-5
- Jacob, D., Elizalde, A., Haensler, A., Hagemann, S., Kumar, P., Podzun, R., et al. (2012). Assessing the transferability of the regional climate model REMO to different coordinated regional climate downscaling experiment (CORDEX) regions. *Atmosphere*, 3(1), 181–199. https://doi. org/10.3390/atmos3010181
- Jones, C., Hughes, J., Bellouin, N., Hardiman, S., Jones, G., Knight, J., et al. (2011). The HadGEM2-ES implementation of CMIP5 centennial simulations. Geoscientific Model Development, 4(3), 543–570. https://doi.org/10.5194/gmd-4-543-2011
- Keuler, K., Radtke, K., Kotlarski, S., & Lüthi, D. (2016). Regional climate change over Europe in COSMO–CLM: Influence of emission scenario and driving global model. *Meteorologische Zeitschrift*, 25(2), 121–136. https://doi.org/10.1127/metz/2016/0662
- Leutwyler, D., Lüthi, D., Ban, N., Fuhrer, O., & Schär, C. (2017). Evaluation of the convection-resolving climate modeling approach on continental scales. *Journal of Geophysical Research: Atmospheres*, 122(10), 5237–5258. https://doi.org/10.1002/2016JD026013
- Marsland, S. J., Haak, H., Jungclaus, J. H., Latif, M., & Röske, F. (2003). The Max-Planck-Institute global ocean/sea ice model with orthogonal curvilinear coordinates. Ocean Modelling, 5(2), 91–127. https://doi.org/10.1016/S1463-5003(02)00015-X
- Mizielinski, M., Roberts, M., Vidale, P., Schiemann, R., Demory, M.-E., Strachan, J., et al. (2014). High-resolution global climate modelling: The UPSCALE project, a large-simulation campaign. *Geoscientific Model Development*, 7(4), 1629–1640. https://doi.org/10.5194/gmd-7-1629-2014
- Nabat, P., Somot, S., Cassou, C., Mallet, M., Michou, M., Bouniol, D., et al. (2020). Modulation of radiative aerosols effects by atmospheric circulation over the Euro-Mediterranean region. *Atmospheric Chemistry and Physics*, 20(14), 8315–8349. https://doi.org/10.5194/acp-20-8315-2020
- Neff, E. L. (1977). How much rain does a rain gage gage? Journal of Hydrology, 35(3–4), 213–220. https://doi.org/10.1016/0022-1694(77) 90001-4
- Powers, J. G., Klemp, J. B., Skamarock, W. C., Davis, C. A., Dudhia, J., Gill, D. O., et al. (2017). The weather research and forecasting model: Overview, system efforts, and future directions. *Bulletin of the American Meteorological Society*, 98(8), 1717–1737. https://doi.org/10.1175/ BAMS-D-15-00308.1
- Raddatz, T., Reick, C., Knorr, W., Kattge, J., Roeckner, E., Schnur, R., et al. (2007). Will the tropical land biosphere dominate the climate–carbon cycle feedback during the twenty-first century? *Climate Dynamics*, 29(6), 565–574. https://doi.org/10.1007/s00382-007-0247-8
- Rockel, B., Will, A., & Hense, A. (2008). The regional climate model COSMO–CLM (CCLM). *Meteorologische Zeitschrift*, 17(4), 347–348. https://doi.org/10.1127/0941-2948/2008/0309
- Skamarock, W. (2008). A description of the advanced research WRF version 3. Tech. Note, 1–96. https://doi.org/10.5065/D68S4MVH
- Tabary, P., Dupuy, P., Lhenaff, G., Gueguen, C., Moulin, L., Laurantin, O., et al. (2012). A 10-year (1997-2006) reanalysis of quantitative precipitation estimation over France: Methodology and first results (pp. 255–260). IAHS-AISH publication.
- van Meijgaard, E., Van Ulft, L., Lenderink, G., De Roode, S., Wipfler, E. L., Boers, R., & van Timmermans, R. (2012). Refinement and application of a regional atmospheric model for climate scenario calculations of Western Europe (No. KVR 054/12). KvR.

- Vautard, R., Kadygrov, N., Iles, C., Boberg, F., Buonomo, E., Bülow, K., et al. (2021). Evaluation of the large EURO-CORDEX regional climate model ensemble. *Journal of Geophysical Research: Atmospheres*, 126(17), e2019JD032344. https://doi.org/10.1029/2019JD032344 Voldoire, A., Sanchez-Gomez, E., y Mélia, D. S., Decharme, B., Cassou, C., Sénési, S., et al. (2013). The CNRM-CM5. 1 global climate model:
- Description and basic evaluation. *Climate Dynamics*, 40(9–10), 2091–2121. https://doi.org/10.1007/s00382-011-1259-y
  - Yang, D., Ishida, S., Goodison, B. E., & Gunther, T. (1999). Bias correction of daily precipitation measurements for Greenland. Journal of Geophysical Research, 104(D6), 6171–6181. https://doi.org/10.1029/1998JD200110